Author’s response to reviews

Title: Repeated mitral valve replacement in a patient with extensive annular calcification

Authors:

Tadashi Kitamura (funcorogash@hotmail.com)
Sachito Fukuda (scfukuda@mitsuihosp.or.jp)
Takahiro Sawada (ipvwx@yahoo.co.jp)
Sumio Miura (smiura-ths@umin.ac.jp)
Ikutaro Kigawa (kigawaik@live.jp)
Takeshi Miyairi (tmiyairi-ky@umin.ac.jp)

Version: 2 Date: 29 October 2011

Author’s response to reviews:

29th October 2011

Mr Vipin Zamvar
Editor-in-Chief
Journal of Cardiothoracic Surgery
c/o BioMed Central
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom

Dear Mr Zamvar

Thank you very much for reviewing our manuscript entitled “Repeated mitral valve replacement in a patient with extensive annular calcification”. I would like to make a response to the reviewers’ comments.

1. Dr Hiroo Takayama

Thanks very much for your positive comments. We really appreciate it.

2. Dr Robert Slight

Thank you very much for your constructive suggestions to make our work acceptable. We referenced the case report mentioned and have made language corrections according to your instruction.

Again, we appreciate your review of this work.

Sincerely,
Tadashi Kitamura MD, PhD
Department of Cardiovascular Surgery
Kitasato University School of Medicine
1-15-1 Kitasato, Minami, Sagamihara
Kanagawa 252-0374
Japan